Author’s response to reviews

Title: An interesting case of likely BRCA2 related bilateral breast cancer with metastasis in the fimbrial part of Fallopian tube

Authors:
Lučka Boltežar (lboltezar@onko-i.si)
Gorana Gašljević (ggasljevic@onko-i.si)
Srdjan Novaković (snovakovic@onko-i.si)
Vida Stegel (vstegel@onko-i.si)
Erik Skof (eskof@onko-i.si)

Version: 1 Date: 06 Mar 2020

Author’s response to reviews:

Response to reviewers

To reviewer 1: The abstract and the discussion were corrected according to your major comment and we hope it is now written more clearly.

To reviewer 2:

Major items:

As you kindly stated in your comment to use the data base provided (https://www.ncbi.nlm.nih.gov/clinvar/variation/38183/), BRCA2 variant c.8755-1G>A is classified as a pathogenic variant so we could not correct our manuscript accordingly (accessed online on 26.2.2020 at 15.00). The origin of the fallopian tube carcinoma was clearly rewritten to be a metastasis of breast carcinoma, due to pathology report.

Minor items:

1. English was improved.

2. Terminology is standardised.

3. Standard abbreviation is used.
4. Spelling is corrected.

5. The sentence is deleted.

6. Corrected.

7. Corrected.

8. Replaced.

9. Replaced.